Author’s response to reviews

Title: Pharmacologic prophylaxis for atrial fibrillation following cardiac surgery: a systematic review

Authors:

Ioanna Koniari (iokoniari@yahoo.gr)
Efstratios Apostolakis (stratisapostolakis@yahoo.gr)
Christina Rogkakou (christinarog@hotmail.com)
Nikolaos G Baikoussis (ngbaik@yahoo.com)
Dimitrios Dougenis (ddougenis@med.upatras.gr)

Version: 5 Date: 1 November 2010

Author’s response to reviews:

dear editor

we created a table as requested. reference 12 is found in the text and in the table as [9-13]. reference 83 is between 82 and 84 in square brackets. reference 61 is also in square brackets. We remove all highlighting from the manuscript. we put the competing interests and author contributions sections are situated before the reference list.

we delete any "et al" from the references.

the authors